Author’s response to reviews

Title: Reproductive health of women with and without disabilities in South India, the SIDE Study (South India Disability Evidence) Study: A Case Control Study

Authors:

Venkata S Murthy Gudlavalleti (gvsmurthy1956@gmail.com)
Neena John (neena.sushil@iiphh.org)
Jayanthi Sagar (jayanthi.sagar@iiphh.org)

Version: 2
Date: 27 June 2014

Author’s response to reviews:

Dear Editors,

We are pleased to upload the revised version of our manuscript based on your suggestions. The changes that have been incorporated in the revised version are:

1. A section on conclusions has now been added. The manuscript is now formatted into the following sections – Background, Methods, Results, Discussion, Conclusions, Competing Interests, Author contributions, Acknowledgements, References and Tables.

2. Line numbers have now been incorporated and all page breaks in the word document have been removed.

3. A statement on competing interests has now been incorporated.

4. Author’s contributions have been included as advised.

5. Acknowledgements have been included.

Kind regards,

Gudlavalleti Venkata S Murthy